

G OPEN ACCESS

Citation: de Groot NS, Torrent Burgas M (2020) Bacteria use structural imperfect mimicry to hijack the host interactome. PLoS Comput Biol 16(12): e1008395. https://doi.org/10.1371/journal. pcbi.1008395

Editor: Arne Elofsson, Stockholm University, SWEDEN

Received: July 7, 2020

Accepted: September 23, 2020

Published: December 4, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1008395

Copyright: © 2020 de Groot, Burgas. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

Funding: Ministerio de Economía y Competitividad (MEC):Marc Torrent RYC-2012-09999; Ministerio

RESEARCH ARTICLE

Bacteria use structural imperfect mimicry to hijack the host interactome

Natalia Sanchez de Groot¹*, Marc Torrent Burgas²*

1 Gene Function and Evolution Lab, Centre for Genomic Regulation (CRG), Dr. Aiguader 88, Barcelona, Spain, **2** Systems Biology of Infection Lab, Department of Biochemistry and Molecular Biology, Biosciences Faculty, Universitat Autònoma de Barcelona, Cerdanyola del Vallès, Spain

* natalia.sanchez@crg.eu (NSdG); marc.torrent@uab.cat (MTB)

Abstract

Bacteria use protein-protein interactions to infect their hosts and hijack fundamental pathways, which ensures their survival and proliferation. Hence, the infectious capacity of the pathogen is closely related to its ability to interact with host proteins. Here, we show that hubs in the host-pathogen interactome are isolated in the pathogen network by adapting the geometry of the interacting interfaces. An imperfect mimicry of the eukaryotic interfaces allows pathogen proteins to actively bind to the host's target while preventing deleterious effects on the pathogen interactome. Understanding how bacteria recognize eukaryotic proteins may pave the way for the rational design of new antibiotic molecules.

Author summary

To fulfil their function, proteins need to interact with each other forming complexes. Understanding how pathogen proteins bind their host counterparts is important to explain how bacteria can infect, survive and proliferate inside cells. To achieve that, pathogen proteins mimic eukaryote interfaces to interact with the host. However, we discovered that such mimicry is imperfect, and pathogen proteins display particular features that are not found in eukaryotic complexes. This imperfect mimicry would allow pathogen proteins to actively bind to the host targets while preventing deleterious effects on the pathogen network. Indeed, we show that highly connected proteins (hubs) in the host-pathogen networks are mostly isolated in the pathogen network. The existence of imperfect mimicry opens the door to the design of new molecules aimed to target host-pathogen complexes with reduced side effects. Hence, in the long term, our results may lay the foundation of a new class of antimicrobials.

Introduction

In nature, bacteria do not exist in isolation but within communities of multiple species that require the bacteria to communicate and organize [1]. In particular, pathogenic bacteria need to interact with host cells to ensure their survival and proliferation [2]. Most of these interactions are mediated by protein effectors that are involved in microbial virulence [3, 4]. These

de Economía y Competitividad (MEC):Marc Torrent SAF2015-72518-EXP; Ministerio de Economía y Competitividad (MEC):Marc Torrent SAF2017-82158-R; European Society of Clinical Microbiology and Infectious Diseases (ESCMID): Marc Torrent ESCMID-2016 The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

effectors, delivered to the host by secretion systems [5], outer membrane vesicles [6] or other specific mechanisms [7], allow bacteria to bind to host cells, replicate and spread within the host and subvert its immune system.

The collection of all these interactions between the host and the pathogen are known as the host-pathogen interactome. In the interactome, proteins with high connectivity degree are known as interactome "hubs" and are associated with essential and conditional phenotypes [8, 9]. The correlation between node degree and gene essentiality is known as the centrality-lethality rule [10] and has been observed in many organisms and interspecies interactomes [11]. In infectious diseases, pathogens selectively target hubs in the host interactome to rewire specific pathways for their own benefit [8]. The elucidation of these interactions is of utmost importance to understand how pathogens hijack the host systems.

Protein interactions are mostly mediated by domain-domain pairs. However, the number of protein folds [12] and interface geometries [13] is limited and these structures are frequently 'reused', allowing a single protein to bind many partners. This recycling strategy is necessary because the interaction space is limited [14] and the distribution of domains in bacteria and eukaryotes has major differences. For example, the most common folds in prokaryotes are those involved in housekeeping functions, such as P-loop-containing NTPases and TIM barrels, whereas the distribution in eukaryotes is dominated by domains with regulatory functions, such as protein kinases and β -propellers [15]. Hence, these constraints set a diversity upper limit and impose restrictions on how bacteria can bind eukaryote proteins.

Despite recent advances in the characterization of host-pathogen interactions, the structural knowledge of these protein complexes is very limited and largely restricts our ability to understand pathological systems. Here, we analyzed the degree centrality of bacterial proteins in the pathogen and the host-pathogen interactomes and investigate the structural characteristics of the interactions involved. We observed that hubs in the host-pathogen interactome are largely isolated in the pathogen interactome. This behavior can be explained by an imperfect mimicry of host interfaces that allow bacteria to minimize the toxicity of these proteins by restricting the number of interactions while maintaining an affinity for host proteins.

Results

Hubs in the host-pathogen interactome are segregated in the pathogen interactome

Because effectors target specific host interactions, their interfaces need to be fine-tuned to interact with a precise set of targets [16]. If proteins are not controlled for promiscuous interactions, this can lead to toxic effects [17]. In the case of effectors, optimized interfaces must be controlled to avoid undesired pathogen self-interactions that may compromise cell performance. A safe ward strategy against unwanted interactions is the timely expression of proteins under certain circumstances [18]. If bacterial effectors were only produced when bacteria are in close contact with the host, their deleterious effects could be minimized. However, this does not seem to be the case. The expression of effectors delivered by secretion systems is not triggered upon infection in cases where data is available (S1A Fig). Also, other mechanisms of delivering virulence factors, such as extracellular vesicles, encapsulate large amounts of proteins from the bacteria cytoplasm and periplasm and do not specifically select their cargo [19]. This strategy would also be non-optimal as detection and killing of bacteria by the innate immune cells is very fast [20] while altering the expression of a protein may take longer [21].

In this context, either these proteins are physically isolated from the rest of the proteome or are integrated into the network in a controlled manner. In eukaryotes, spurious interactions are prevented by segregating proteins into different compartments. However, bacteria lack

these compartments. A possible alternative would be the use of protein condensates [22], nonmembrane bound structures formed by liquid-liquid phase separation [23]. These condensates would allow proteins to be relatively isolated from the cell milieu, but ready to be delivered when required. However, we did not find a clear difference in condensate propensity between secreted effectors and the rest of the proteome (S1B Fig). Besides, bacteria are more proficient in forming solid aggregates, that are less dynamic than liquid condensates [24, 25].

Based on these evidences, we hypothesized that effectors could be integrated into the bacteria network in such a way that they were not deleterious when pathogens replicate outside the host but can be effectively deployed upon infection. By isolating effectors from the pathogen network, they would have less control over the interactome, minimizing the side effects. In agreement with this reasoning, we observed that effector proteins are significantly depleted of pathogen hubs (**Fig 1A**). These results were also validated in an orthogonal database to control for potential methodological bias (**Fig 1A**, see <u>Materials and Methods</u> for a detailed description of the control database).

To further investigate the integration of effectors in protein networks we compared its degree centrality both in the pathogen and the host-pathogen interactomes (**Fig 1B**). We were able to separate proteins into three clusters: (*i*) proteins that have a high number of interactions in the host-pathogen interactome but are highly isolated in the pathogen interactome (C1 cluster, **Fig 1C**), (*ii*) proteins that are isolated in the host-pathogen interactome but largely connected in the pathogen interactome (C2 cluster, **Fig 1C**) and (*iii*) proteins that are mainly isolated in both interactomes (C3 cluster, **Fig 1C**).

We investigated whether the structural properties of these proteins may explain the differences observed between clusters (**Fig 1D**). We found that the proteins belonging to the C1 cluster are richer in coil structure compared to proteins in clusters C2 and C3. This is consistent with an increased propensity to disordered regions, a property commonly observed in eukaryotic proteins. The proteins that define the C2 cluster tend to form alpha helix, which favors the binding to nucleic acids. Finally, the proteins belonging to the C3 cluster are enriched in beta-sheet structure and more prone to aggregate. Overall, these clusters show specific structural properties and may reflect the differences in the cellular milieu between prokaryotes and eukaryotes.

The fact that we did not find any cluster with a high number of interactions in both interactomes suggests that protein interfaces in the host and the pathogen are nearly orthogonal, meaning that is difficult to optimize an interface to act as a hub in the pathogen and the hostpathogen interactomes at the same time. These observations raise a fundamental question: how can bacteria discriminate between self and non-self interfaces?

Bacteria use structural imperfect mimicry to interact with the host

Protein interactions are mediated by non-covalent bonds between residues located in the interaction core, which is a central area excluded from the solvent. This core region is surrounded by the rim area, which is partially buried, helps to exclude water molecules from the core and is involved in the modulation of the interaction. The core explains most of the binding energy of the complex, while the rim can tune the binding strength [26], particularly in small complexes [27].

To understand how the interface structure can be used to discriminate self and non-self interfaces in bacteria, we mined the PDB for bacteria-eukaryote (BE) complexes and found 89 nonredundant entries (Fig 2A, S1 Table). The BE dataset used is enriched in pathogenic proteins such as effectors, toxins and adhesins (S1 Table). Bacterial proteins included in the BE set are significantly upregulated in infection and are relevant for pathogenesis (S2 Fig). We



Fig 1. Hubs in the host-pathogen interactome are largely isolated in the pathogen network. (A) We computed the degree of interaction for all proteins in *Y. pestis*, *B. anthracis* and *F. tularensis* and asked whether the number of hubs (here defined as the 5% of most connected proteins) differ between effectors and non-effectors. In all cases, the number of hubs observed is significantly lower than expected. Similar results were obtained in the control database. The p-values were computed using a χ^2 -square test of independence to assess the probability of observing such a large discrepancy (or larger) between observed and expected values. Effectors were defined using EffectiveDB with a stringency threshold of 0.95 [52]. All comparisons remain statistically significant in the threshold interval 0.90–0.99 (p<0.05). (B) We compared the degree centrality of bacterial proteins in the pathogen and the host-pathogen interactomes. Based on the results obtained, we classified the bacterial proteins in clusters,

according a k-means clustering algorithm. (C) Based on this clustering, three different groups were identified: proteins that have a high number of interactions in the host-pathogen interactome but are highly isolated in the pathogen interactome (C1 cluster), proteins that are isolated in the host-pathogen interactome but deeply connected in the pathogen interactome (C2 cluster) and proteins that are mainly isolated in both clusters (C3 cluster). (D) The three clusters identified previously have distinctive structural properties. Proteins in C1 cluster are enriched in coil structure, which favors the presence of disordered regions; proteins in C2 cluster are enriched in alpha helix structure, which favors interaction with nucleic acids and proteins in C3 cluster are enriched in beta sheet structure, that favors aggregation. * p<0.05; ** p<0.01; ***p<0.005 using a Mann–Whitney U-test with $\alpha = 0.05$.

https://doi.org/10.1371/journal.pcbi.1008395.g001

also selected 183 bacteria-bacteria (BB) and 686 eukaryote-eukaryote (EE) complexes for comparison (S1 Table). We divided proteins into three regions (interface, rim and surface) and analyzed the amino acid composition for each region. Overall, we could clearly distinguish these three regions based on their composition, with polar residues favored at the surface and hydrophobic residues more abundant in rim and interface regions (Fig 2B). Differences in each of these regions between BE, EE and BB complexes were more subtle (Fig 2B and 2C). While some residues (Trp, Phe and Lys) were enriched in the rim area in BE complexes, only differences in Leu composition were detected in the interface area (S3 and S4 Figs). This fact may be related to the higher conservation of the interface compared to the rim or surface regions [28, 29]. It is also important to note that "affinity-defining" positions, located in the interface, are highly optimized whereas "specificity-defining" positions are usually non-optimal and are located at the rim area [16]. Hence, bacteria proteins may preferentially modify the rim area to discriminate between self and non-self interactions.

Despite maintaining the same composition at the interface level, the interaction pattern between amino acids was substantially different between complex types. We evaluated aminoacid interactions and compared the connectivity network of BE complexes with that of EE and BB complexes. In general, we found that the contribution, in terms of the number of interactions, for each amino acid in BE complexes was significantly correlated to EE but not to BB complexes (Fig 2D). The correlation between the network of interactions for each amino acid (Fig 2E and 2F) and their organization (S5 Fig) also confirmed that BE complexes were more similar to EE than BB complexes. Random resampling confirmed that our measurements were not affected by sample size (Fig 2F, insert). These results support the theory that bacteria may use molecular mimicry to interact with host proteins. According to the mimicry hypothesis, bacteria can partly or completely imitate the structure of host proteins by mechanisms of gene transfer and/or convergent evolution using a strategy called 'molecular mimicry' [30, 31]. Bacterial proteins competitively bind to the target host site [32, 33] and redirect host hub proteins away from their pathway [34, 35]. This strategy does not necessarily involve changing the entire structure of proteins but only certain residues in the interface or rim areas [34, 35]. These bacterial proteins target host processes involved in cell adherence and invasion, which are essential for infection and explain why certain bacteria display strict host selectivity [36]. However, mimicry has been observed mostly on a case-by-case basis, using sequence or structure similarity [2, 37] or by solving isolated complexes [38].

While the evidence supporting structural mimicry is strong, we noticed clear differences between BE and EE complexes at the amino acid interaction level (Fig 2E). For example, Arg interactions had different preferences: Arg-Glu interactions were preferred in BE complexes, whereas Arg-Asp interactions were preferred in EE complexes. This might reflect an evolutionary adaptation, as Glu residues are preferred in eukaryotic interfaces compared to prokaryotic interfaces and vice versa for Asp residues (S3 Fig, p = 0.016). In these lines, when analyzing directionality in BE interactions, we observed that some amino acids were frequently targeted at the bacterial interface (Tyr, Arg, Leu and Gly), whereas others were mainly targeted at the eukaryotic interface (Trp, Lys and Met; Fig 2G), being Trp was the most conspicuous case (S6 Fig). We noticed that Trp-Asp and Trp-Glu interactions were more common in BE



Fig 2. Analysis of protein complexes. (**A**), To compare the structural determinants of bacteria-eukaryote (BE) complexes, 89 nonredundant complexes were obtained from the PDB and compared to 183 bacteria-bacteria (BB) and 686 eukaryote-eukaryote (EE) complexes. (**B**) Hierarchical clustering and (**C**), principal component analysis of amino acid composition in BE, BB and EE protein complexes. All percentages were controlled by the amino acid structure of the protein. (**D**) To characterize the interaction pattern in BE complexes, the number of interactions for each amino acid in BE complexes was plotted against EE and BB complexes. Regression lines were calculated using the Spearman rank-correlation approach to control for the impact of extreme values. (**E**) Hierarchical clustering of the interaction pattern for each amino acid. The correlation coefficient for each amino acid was calculated comparing the number of interactions with all other amino acids in BE complexes. The inner panel shows the p-value distribution in a resampling control to correct for effect size (see <u>Materials and Methods</u>) (**G**) Directionality for each amino acid (D_i) in BE interactions was calculated as the relative difference in the number of interactions (*N*) in both directions $D_i = \frac{N_i^B - N_i^E}{N_i^B + N_i^E}$. (**H**) Distribution of the angle measured for all anion-pi interactions involving

Trp in BE complexes.

https://doi.org/10.1371/journal.pcbi.1008395.g002

complexes (25% of proteins had at least 1 anion-pi interaction) compared to the PDB interactome (less than 10% of proteins had at least 1 interaction) [39]. Asp and Glu were preferred in the bacterial interface, while Trp was mostly located in the eukaryotic interface, which coincides with Trp being more abundant in eukaryotes than bacteria (p-values 0.10 and 0.012, for core and rim, respectively). Furthermore, Trp in the eukaryotic interface had a higher contribution to complex stability compared to Trp in the prokaryotic interface (S7 Fig), suggesting that those interactions would contribute to complex stability. In almost all interactions in BE complexes (95%), Trp interacted with anionic residues through anion-pi interactions, which involves the contact of the negative density of Asp and Glu with the positive density at the edge of the aromatic ring (Fig 2H). Collectively, these results confirm that bacteria use molecular mimicry to interact with eukaryotic proteins, but also suggest that such mimicry is imperfect. Hence, although the composition of the central interface is similar across all complexes, the differences observed in its geometry can help discriminate between self and non-self interactions. Also, differences in the rim area would allow to fine-tune the binding. In the next section, we explore the use of imperfect mimicry in the context of host-pathogen interactions.

Imperfect mimicry in the Y. pestis-H. sapiens interactome

During the course of infection, pathogens use proteins to rewire a myriad of biochemical processes [40] that are required for efficient propagation [41, 42]. We recently showed that pathogen proteins engaged in a higher number of interactions with the host also have a major impact on pathogen fitness during infection [8]. Hence, the relevance of pathogen proteins in the infection process is proportional to its ability to reorganize the host interactome. Unfortunately, complexes of bacterial proteins with human targets are largely underrepresented in the PDB database.

To further investigate this issue, we used the *Yersinia pestis-Homo sapiens* interactome and analyzed domain-domain associations (in terms of protein superfamilies) in comparison with the isolated *Y. pestis* and *H. sapiens* networks. Such an approach is justified because organisms mainly use the same 'building blocks' for protein interactions, and the function of domain pairs seem to be maintained during evolution [43, 44]. We observed that an important number of associations are shared between the *Y. pestis-H. sapiens* interactome and the *H. sapiens* interactome (19%) compared with the *Y. pestis* interactome (0.72%, p<0.00001; **Fig 3A**). Consistently, the shared subnetwork (intersection) between BE and EE networks is more densely connected compared to the shared subnetwork between BE and BB networks (**Fig 3B and 3C**). Again, this suggests that the BE interactome is more closely related to the EE rather than the BB interactome.

To further validate these results, we filtered the *Y*. *pestis-H*. *sapiens* network with fitness data, which measures the relevance of a given bacterial gene in infection. Using this strategy, we created a subset of domain interactions that have a high impact on the fitness of *Y*. *pestis*



Fig 3. Structural analysis of protein-protein interactions in the *Yersinia pestis-Homo sapiens* interactome. (A) Venn diagram showing shared and unique domain associations between the *Y. pestis-H. sapiens*, *Y. pestis* and *H. sapiens-H. sapiens* interactomes. (B) Percentage of isolated and connected nodes in the shared subnetworks (intersection) between the *Y. pestis-H. sapiens* interactome and the *Y. pestis-Y. pestis* or *H. sapiens-H. sapiens* interactomes. Cumulative distribution of betweenness centrality in both subnetworks. (C) *Y. pestis-H. sapiens* domain association network filtered for *Y. pestis* proteins that have a high contribution to infection fitness (fitness factor < 0.4). Complexes that involve bacterial proteins with a high contribution to infection fitness were modeled and docked to obtain the putative three-dimensional structure. (D) Distribution of Pearson correlation coefficients in the filtered network of contacts for all modeled complexes (n = 18). For all amino acids the connectivity matrix and Pearson correlation coefficients were calculated. The plot shows the distribution of correlation coefficients for each amino acid in YH complexes against HH and YY complexes.

https://doi.org/10.1371/journal.pcbi.1008395.g003

during infection. The superfamily associations for such network are significantly enriched in domains related to infection (Fig 3C, S2 Table). When possible, we modelled the three-dimensional structure of the proteins involved in this network by sequence similarity and then obtained the structure of the complex by docking simulations (18 complexes). Docking procedures were not highly reliable to delineate interfaces in detail but helped to draw a coarsegrained view of the interactions. To investigate whether the predicted complexes are more similar to BB or EE complexes, the correlation coefficients for the interaction pattern of each residue were obtained (Fig 3D). Similar to previous results, we observed that the modelled interactions were more similar to EE complexes than BB complexes (Fig 2F). Overall, the correlation coefficients were lower when compared to those of the complexes deposited in the PDB, which can be attributed to the predicted nature of these complexes. Hence, although modelled data must be treated with caution, it reflects a general trend that is consistent with previous observations.

Discussion

Based on the results presented here, we suggest that bacterial effectors have evolved their interfaces to imperfectly mimic eukaryotic complexes. During this process, effectors would have been subjected to two opposing forces: on the one hand, there would be an evolution-ary pressure to increase the number of interactions with the host while, on the other hand, effectors would be forced to minimize the number of interactions with other pathogen proteins (Fig 4A). The first condition is necessary to increase the pathogen infectivity and its survival inside the host cells. The second one is less obvious but can be explained in terms of protein stickiness, which is higher in eukaryotes than prokaryotes. By mimicking eukaryotic interfaces, effectors become stickier, potentially increasing the number of spurious interactions with other pathogen proteins. Such poisonous interactions may compromise the cell viability; therefore, pathogens must find a balance between increasing infectivity and limiting toxicity. Using imperfect mimicry of eukaryotic interfaces, effectors are able to discriminate between bacteria-bacteria (self) and bacteria-host (non-self) interactions (Fig 4B).

The imperfect mimicry of protein interfaces has direct evolutionary consequences: as pathogen effectors mimic eukaryotic complexes to enhance adaptation, the host, in its turn, evolves its proteome to discriminate its own proteins from the pathogen mimics. This creates an arms race for survival between the host and the pathogen. Such behavior is observed in viral infections, particularly in those caused by poxviruses [45]. In a viral infection, host cells phosphorylate the eukaryotic initiation factor 2A (eIF2a) by protein kinase R (PKR) to inactivate translation. To restore translation, poxviruses evolved a protein called K3L that mimics eIF2a and competes with it for PKR phosphorylation. In its turn, primates also evolve eIF2a to discriminate it from K3L, creating a lasting evolutionary circle.

Effectors regulate pathogen adhesion, survival and proliferation in the host and so, they are frequently found to be essential for infection *in vivo* [46]. However, as mentioned before, these proteins are also isolated within the pathogen interactome, which explains why they are often classified as nonessential for the pathogen growth *in vitro* [8]. Unfortunately, most large-scale screening assays aimed to discover new antimicrobials are developed *in vitro*. Using such approaches, most effectors will never be discovered and the potential drugs that could be developed against them will remain unexplored. Our observations, therefore, confirm the need to redefine our discovery pipelines to properly reflect the host environment.

Last but not least, our results suggest that host-pathogen protein-protein interactions are potential targets for a new generation of antimicrobials. If an interaction is required for the pathogen to infect the host, blocking this interaction would help to stop or delay the infection. Hence, compounds designed to interfere with such host-pathogen interactions would behave as antimicrobials. This idea has been suggested by us previously but also by other authors. The results presented in this paper conceptually takes us one step further in this effort. We propose that bacteria use imperfect mimicry to interact with host proteins. In other words, the network of interactions in host-pathogen PPIs is different from the own host network. Because of this, we could think of compounds, either peptidomimetics or small molecules, that interact with the bacteria interface without disrupting the host complexes. Our observations suggest that such compounds might have therapeutic potential as the interaction with their host targets would be weaker compared with their bacteria counterparts. In summary, treatments interfering with the adhesion and invasion of bacteria to host cells could be used as preventive strategies during surgical procedures or after infection by reducing the resistance of pathogens to known antibiotics by combating their spread in the organism [47].





https://doi.org/10.1371/journal.pcbi.1008395.g004

Materials and methods

Databases

The *H. sapiens-Y. pestis*, *H. sapiens-B. anthracis* and *H. sapiens-F. tularensis* interactomes were obtained by yeast-two-hybrid (Y2H) and downloaded from IntAct as reported in the study performed by Dyer et al. [48]. In Y2H, cells produce high amounts of proteins that can lead to

spurious interactions (false positives). Also, Y2H can fail to detect some interactions due to limitations of the screening (false negatives). The latter can happen when protein fusion disrupts the interaction interface or interferes with protein folding and when proteins fail to localize to the nucleus. Also, in host-pathogen interactions, both proteins must be present in the same cellular location to interact. This can generate a detection bias: some interactions may not be biologically significant while others can remain undetected. To control for biases in the Y2H, a control dataset was built using the PHISTO database: http://www.phisto.org [49], including the following organisms: *Escherichia coli, Pseudomonas aeruginosa, Salmonella enterica* and *Shigella flexneri*. The interactions in the control database were defined using different methodological approaches, including pull-down, affinity coimmunoprecipitation and affinity chromatography, among others.

Bacteria-eukaryote (BE, 89 complexes), bacteria-bacteria (BB, 183 complexes) and eukaryote-eukaryote (EE, 686 complexes) were obtained from the Protein Data Bank (PDB). The codes for all proteins selected in the study are included in <u>S1 Table</u>. We acknowledge that the PDB is biased by (i) the preferences of individual investigators and (ii) the physicochemical nature of proteins, which defines the probability of proteins to crystalize. These reasons may explain the higher abundance of EE complexes in the PDB compared to EB and BB complexes. Differences in size between proteins deposited in the PDB and the corresponding genomes have been also reported [50]. Despite these biases, the interface size, measured as the number of interactions per molecule, in all complexes included in our dataset is similar (differences not significant, p > 0.05 using a Mann–Whitney U-test).

The *Y. pestis* and *H. sapiens* interactomes were obtained from the String database [51]. Only highly reliable interactions were included (experimentally validated interactions with score > 700). The *Y. pestis* interactome has 3701 interactions for 3973 distinct protein-coding genes and the *H. sapiens* interactome has 18982 interactions for 19566 distinct protein-coding genes. Hence, the number of interactions per protein is similar and we do not expect a bias related to relative size.

EffectiveDB (http://effectors.org/) was used to classify proteins between effectors and noneffectors, using a stringent value of 0.95 [52]. Fitness values were obtained from [53] using transposon sequencing (Tn-seq) and calculated as the ratio of the rates of population expansion for the two genotypes after infection of *Y. pestis* in a mouse model. In total, 1.5 million independent insertion mutants were screened with a coverage of ~70% of the *Y. pestis* genome. Protein superfamilies of *Y. pestis* and *H. sapiens* were obtained from UniProt. Structural parameters were obtained from [54] (alpha helix, beta-sheet and coil propensity), [55] (aggregation propensity), [56] (disorder propensity) and [57] (nucleic acid binding propensity).

Interface definition and calculation of contact maps

Residue Interaction Networks (RINs) represent amino acid residues as nodes in a network. If two residues interact (based on spatial distance) they are connected by edges between them. Hence, we decided to compare the connectivity profiles of protein complexes and used them as interaction fingerprints. The interface, rim and surface regions were defined using a python script developed by the Oxford Protein Informatics group, which is freely available (http:// www.stats.ox.ac.uk/~krawczyk/GetContacts.zip). Briefly, the interface residues were defined as those in close contact between two molecules in a given complex (4.5 Å). Rim residues were not engaged in intermolecular contacts but were close to the interface (contact between molecules < 10 Å) and can have a more subtle effect on binding. Surface residues were determined as residues not present in either the rim or interface region that display a surface accessible area greater than 20 Å². We considered that Trp residues were engaged in anion-pi interactions when the distance between the centroid of the aromatic ring and the anion was between 2-5 Å. Anion-pi distances and interaction angles (defined between 0° and 90°) were measured in Pymol. Contact maps were generated in R (version 3.4.4) using the function *cmap* included in the bio3d package [58].

Modelling and docking of Y. pestis-H. sapiens complexes

Proteins involved in complexes were retrieved from the PDB when possible. Otherwise, the three-dimensional structure was modelled using Modeller (version 9.21) [59] as long as a template had homology higher than 30%, spanning more than 75% of the protein length. Docking was performed using Frodock (version 2.0) with default parameters [60]. The interface for complexes ranked highest in the docking score was analyzed using the pipeline described before.

Network analysis

All protein networks were analyzed with Cytoscape (version 3.6.1) [61], and statistical calculations were performed in R (version 3.4.4). The degree (k) of a node *i* is defined as the number of edges linked to *i*. To compare the node degree between two networks, we define the normalized degree as k/(n-1) where n is the number of nodes in the network. Betweenness centrality (C_h) was computed as follows:

$$C_b(i) = \sum_{s \neq i \neq t} \frac{\sigma_{st}(i)}{\sigma_{st}}$$

where *s* and *t* are nodes in the network different from *i*, σ_{st} denotes the number of shortest paths from s to t, and $\sigma_{st}(i)$ is the number of shortest paths from s to t that i lies on. The intersection of two networks was calculated using the merge function (intersection) in Cytoscape.

Statistical analyses

Unless otherwise specified, all p-values were calculated using the Mann-Whitney U-test and considered significant when p<0.05 (see Figure Legends for further details). The χ^2 test was used to determine whether there is a significant difference between the expected and the observed frequencies in two categories. In all cases, two-sided tests were used with a testing level $\alpha = 0.05$. Random resampling was used to create balanced datasets and determine the effect of sample size. To generate balanced datasets, the majority class was randomly undersampled while the minority class was oversampled. Using this strategy, 500 balanced datasets were generated to assess the effect size.

Supporting information

S1 Fig. Strategies to control potentially detrimental interactions in the pathogen interactome. Pathogens could (A) regulate the expression of protein effectors or (B) compartmentalize the interactions in protein condensates for timely delivery upon infection. As observed in the boxplots in the lower panel, no clear significant differences were observed in all three organisms studied. P-values were calculated using the Mann-Whitney U test for comparing pairs of independent samples.

(TIFF)

S2 Fig. Fitness impact and differential expression of bacterial proteins belonging to the BE set. To validate the role of bacterial proteins in the BE dataset during infection, we used blastp against the BacFitBase database (http://www.tartaglialab.com/bacfitbase/) to predict the fitness score associated with each protein. The fitness defect associated to these proteins is larger than a null effect (p = 0.0005, negative values indicate a reduction on pathogen fitness), suggesting that the BE dataset is associated with pathogenesis. Also, we analyzed the mRNA expression data using the DualSeqDB database (http://www.tartaglialab.com/dualseq/). Proteins belonging to the BE dataset are significantly upregulated in infection (p<0.0001) suggesting its implication in bacterial pathogenesis.



S3 Fig. Amino acid composition of interface, rim and surface regions in bacteria-eukaryote complexes. Percentage of each amino acid in the interface, rim and surface regions of bacteria-eukaryote complexes. Each region is subdivided in bacteria (B, red) and eukaryote counterparts (E, blue).

(TIFF)

S4 Fig. Amino acid preference in interface, rim and surface regions of bacteria-eukaryote complexes. Venn diagram displaying the differential amino acid composition of interface, rim and surface areas in bacteria-eukaryote complexes. Significant amino acid composition differences between bacteria and eukaryote counterparts in bacteria-eukaryote complexes were calculated using a Mann-Whitney U test and considered significant when p < 0.05. (TIFF)

S5 Fig. Comparison of interaction patterns in bacteria-eukaryote complexes compared to bacteria and eukaryote complexes. Dendrograms for each complex type (bacteria, eukaryote and bacteria-eukaryote) were built according to the interaction pattern of each amino acid using the Ward's minimum variance method. For each dendrogram, four groups (red, green, orange and blue) were defined using k-means clustering. Then, tanglegrams were built to compare the congruence between dendrograms (bacteria-eukaryote compared to bacteria and eukaryote). In the figure, congruence is depicted by the number of colored lines mapping common elements between same groups in different dendrograms. The quality of the alignment of the two dendrograms (entanglement) is also reported as a quantitative measure of congruence. Entanglement is measured between 1 (full entanglement, high congruence) to 0 (no entanglement, null congruence). Tanglegrams were built using the *dendextend* package in R. (TIFF)

S6 Fig. Amino acid connectivity analysis of bacteria-eukaryote, bacteria and eukaryote complexes. Number of interactions for each amino acid (measured as the percentage relative to the total number of interactions) in bacteria-eukaryote (BE, red), bacteria (BB, blue) and eukaryote (EE, green) complexes. Each pair of letters correspond to an amino acid in the interaction pair. The scale measures the percentage of a given interaction over all detected interactions.

(TIFF)

S7 Fig. Residue contribution to the complex stability by alanine scanning in bacteriaeukaryote complexes. Each residue in the interface of bacteria-eukaryote complexes was mutated to alanine and the change in complex stability was calculated using FoldX. The impact in complex stability for mutations in bacteria (red) and eukaryote (blue) interaction counterparts is compared.

(TIFF)

S1 Table. Uniprot codes of protein structures included in this study. (XLSX)

S2 Table. List of superfamily associations significantly enriched in domains related to infection.

(XLSX)

S3 Table. List of cluster associations related to Fig 1C. (XLSX)

Author Contributions

Conceptualization: Natalia Sanchez de Groot, Marc Torrent Burgas.

Formal analysis: Natalia Sanchez de Groot.

Funding acquisition: Marc Torrent Burgas.

Methodology: Natalia Sanchez de Groot, Marc Torrent Burgas.

Supervision: Marc Torrent Burgas.

Writing - original draft: Natalia Sanchez de Groot, Marc Torrent Burgas.

Writing - review & editing: Natalia Sanchez de Groot, Marc Torrent Burgas.

References

- Ratzke C, Gore J. Self-organized patchiness facilitates survival in a cooperatively growing Bacillus subtilis population. Nat Microbiol. 2016; 1:16022. https://doi.org/10.1038/nmicrobiol.2016.22 PMID: 27572641
- Drayman N, Glick Y, Ben-nun-shaul O, Zer H, Zlotnick A, Gerber D, et al. Pathogens use structural mimicry of native host ligands as a mechanism for host receptor engagement. Cell Host Microbe. 2013; 14(1):63–73. https://doi.org/10.1016/j.chom.2013.05.005 PMID: 23870314
- Yang H, Ke Y, Wang J, Tan Y, Myeni SK, Li D, et al. Insight into bacterial virulence mechanisms against host immune response via the Yersinia pestis-human protein-protein interaction network. Infect Immun. 2011; 79(11):4413–24. https://doi.org/10.1128/IAI.05622-11 PMID: 21911467
- Memisevic V, Zavaljevski N, Pieper R, Rajagopala SV, Kwon K, Townsend K, et al. Novel Burkholderia mallei virulence factors linked to specific host-pathogen protein interactions. Mol Cell Proteomics. 2013; 12(11):3036–51. https://doi.org/10.1074/mcp.M113.029041 PMID: 23800426
- Costa TR, Felisberto-Rodrigues C, Meir A, Prevost MS, Redzej A, Trokter M, et al. Secretion systems in Gram-negative bacteria: structural and mechanistic insights. Nat Rev Microbiol. 2015; 13(6):343–59. https://doi.org/10.1038/nrmicro3456 PMID: 25978706
- Guerrero-Mandujano A, Hernandez-Cortez C, Ibarra JA, Castro-Escarpulli G. The outer membrane vesicles: Secretion system type zero. Traffic. 2017; 18(7):425–32. https://doi.org/10.1111/tra.12488 PMID: 28421662
- Guiral S, Mitchell TJ, Martin B, Claverys JP. Competence-programmed predation of noncompetent cells in the human pathogen Streptococcus pneumoniae: genetic requirements. Proc Natl Acad Sci U S A. 2005; 102(24):8710–5. https://doi.org/10.1073/pnas.0500879102 PMID: 15928084
- Crua Asensio N, Munoz Giner E, de Groot NS, Torrent Burgas M. Centrality in the host-pathogen interactome is associated with pathogen fitness during infection. Nat Commun. 2017; 8:14092. https://doi. org/10.1038/ncomms14092 PMID: 28090086
- Ahmed H, Howton TC, Sun Y, Weinberger N, Belkhadir Y, Mukhtar MS. Network biology discovers pathogen contact points in host protein-protein interactomes. Nat Commun. 2018; 9(1):2312. <u>https:// doi.org/10.1038/s41467-018-04632-8 PMID: 29899369</u>
- Jeong H, Mason SP, Barabasi AL, Oltvai ZN. Lethality and centrality in protein networks. Nature. 2001; 411(6833):41–2. https://doi.org/10.1038/35075138 PMID: 11333967
- Raman K, Damaraju N, Joshi GK. The organisational structure of protein networks: revisiting the centrality-lethality hypothesis. Syst Synth Biol. 2014; 8(1):73–81. <u>https://doi.org/10.1007/s11693-013-9123-5</u> PMID: 24592293

- Chothia C. Proteins. One thousand families for the molecular biologist. Nature. 1992; 357(6379):543–4. https://doi.org/10.1038/357543a0 PMID: 1608464
- Gao M, Skolnick J. Structural space of protein-protein interfaces is degenerate, close to complete, and highly connected. Proc Natl Acad Sci U S A. 2010; 107(52):22517–22. https://doi.org/10.1073/pnas. 1012820107 PMID: 21149688
- Aloy P, Russell RB. Ten thousand interactions for the molecular biologist. Nat Biotechnol. 2004; 22 (10):1317–21. https://doi.org/10.1038/nbt1018 PMID: 15470473
- Wolf YI, Brenner SE, Bash PA, Koonin EV. Distribution of protein folds in the three superkingdoms of life. Genome Res. 1999; 9(1):17–26. PMID: 9927481
- Fromer M, Shifman JM. Tradeoff between stability and multispecificity in the design of promiscuous proteins. PLoS Comput Biol. 2009; 5(12):e1000627. https://doi.org/10.1371/journal.pcbi.1000627 PMID: 20041208
- Vavouri T, Semple JI, Garcia-Verdugo R, Lehner B. Intrinsic protein disorder and interaction promiscuity are widely associated with dosage sensitivity. Cell. 2009; 138(1):198–208. <u>https://doi.org/10.1016/j.cell.</u> 2009.04.029 PMID: 19596244
- Ge H, Liu Z, Church GM, Vidal M. Correlation between transcriptome and interactome mapping data from Saccharomyces cerevisiae. Nat Genet. 2001; 29(4):482–6. <u>https://doi.org/10.1038/ng776</u> PMID: 11694880
- Kuehn MJ, Kesty NC. Bacterial outer membrane vesicles and the host-pathogen interaction. Genes Dev. 2005; 19(22):2645–55. https://doi.org/10.1101/gad.1299905 PMID: 16291643
- Davies WA. Kinetics of killing Listeria monocytogenes by macrophages: rapid killing accompanying phagocytosis. J Reticuloendothel Soc. 1983; 34(2):131–41. PMID: 6411918
- Li GW, Burkhardt D, Gross C, Weissman JS. Quantifying absolute protein synthesis rates reveals principles underlying allocation of cellular resources. Cell. 2014; 157(3):624–35. <u>https://doi.org/10.1016/j.cell.2014.02.033</u> PMID: 24766808
- Abbondanzieri EA, Meyer AS. More than just a phase: the search for membraneless organelles in the bacterial cytoplasm. Curr Genet. 2019; 65(3):691–4. <u>https://doi.org/10.1007/s00294-018-00927-x</u> PMID: 30603876
- 23. Sanchez de Groot N, Torrent Burgas M, Ravarani CN, Trusina A, Ventura S, Babu MM. The fitness cost and benefit of phase-separated protein deposits. Mol Syst Biol. 2019; 15(4):e8075. <u>https://doi.org/10.15252/msb.20178075 PMID: 30962358</u>
- Torrent M, Pulido D, Nogues MV, Boix E. Exploring new biological functions of amyloids: bacteria cell agglutination mediated by host protein aggregation. PLoS Pathog. 2012; 8(11):e1003005. https://doi. org/10.1371/journal.ppat.1003005 PMID: 23133388
- 25. Schramm FD, Schroeder K, Jonas K. Protein aggregation in bacteria. FEMS Microbiol Rev. 2019; 44 (1):54–72.
- Brender JR, Zhang Y. Predicting the Effect of Mutations on Protein-Protein Binding Interactions through Structure-Based Interface Profiles. PLoS Comput Biol. 2015; 11(10):e1004494. https://doi.org/10.1371/ journal.pcbi.1004494 PMID: 26506533
- Agius R, Torchala M, Moal IH, Fernandez-Recio J, Bates PA. Characterizing changes in the rate of protein-protein dissociation upon interface mutation using hotspot energy and organization. PLoS Comput Biol. 2013; 9(9):e1003216. https://doi.org/10.1371/journal.pcbi.1003216 PMID: 24039569
- Guharoy M, Chakrabarti P. Conservation and relative importance of residues across protein-protein interfaces. Proc Natl Acad Sci U S A. 2005; 102(43):15447–52. https://doi.org/10.1073/pnas. 0505425102 PMID: 16221766
- Teppa E, Zea DJ, Marino-Buslje C. Protein-protein interactions leave evolutionary footprints: High molecular coevolution at the core of interfaces. Protein Sci. 2017; 26(12):2438–44. <u>https://doi.org/10.1002/pro.3318 PMID: 28980349</u>
- Sikora S, Strongin A, Godzik A. Convergent evolution as a mechanism for pathogenic adaptation. Trends Microbiol. 2005; 13(11):522–7. https://doi.org/10.1016/j.tim.2005.08.010 PMID: 16153847
- Elde NC, Malik HS. The evolutionary conundrum of pathogen mimicry. Nat Rev Microbiol. 2009; 7 (11):787–97. https://doi.org/10.1038/nrmicro2222 PMID: 19806153
- 32. Stebbins CE, Galan JE. Structural mimicry in bacterial virulence. Nature. 2001; 412(6848):701–5. https://doi.org/10.1038/35089000 PMID: 11507631
- **33.** Escoll P, Mondino S, Rolando M, Buchrieser C. Targeting of host organelles by pathogenic bacteria: a sophisticated subversion strategy. Nat Rev Microbiol. 2016; 14(1):5–19. <u>https://doi.org/10.1038/</u> nrmicro.2015.1 PMID: 26594043

- Guven-Maiorov E, Tsai CJ, Nussinov R. Pathogen mimicry of host protein-protein interfaces modulates immunity. Semin Cell Dev Biol. 2016; 58:136–45. <u>https://doi.org/10.1016/j.semcdb.2016.06.004</u> PMID: 27287306
- **35.** Via A, Uyar B, Brun C, Zanzoni A. How pathogens use linear motifs to perturb host cell networks. Trends Biochem Sci. 2015; 40(1):36–48. https://doi.org/10.1016/j.tibs.2014.11.001 PMID: 25475989
- Pan X, Yang Y, Zhang JR. Molecular basis of host specificity in human pathogenic bacteria. Emerg Microbes Infect. 2014; 3(3):e23. https://doi.org/10.1038/emi.2014.23 PMID: 26038515
- **37.** Doxey AC, McConkey BJ. Prediction of molecular mimicry candidates in human pathogenic bacteria. Virulence. 2013; 4(6):453–66. https://doi.org/10.4161/viru.25180 PMID: 23715053
- Lin DY, Diao J, Chen J. Crystal structures of two bacterial HECT-like E3 ligases in complex with a human E2 reveal atomic details of pathogen-host interactions. Proc Natl Acad Sci U S A. 2012; 109 (6):1925–30. https://doi.org/10.1073/pnas.1115025109 PMID: 22308380
- Lucas X, Bauza A, Frontera A, Quinonero D. A thorough anion-pi interaction study in biomolecules: on the importance of cooperativity effects. Chem Sci. 2016; 7(2):1038–50. <u>https://doi.org/10.1039/ c5sc01386k PMID: 29899893</u>
- 40. de Groot NS, Torrent Burgas M. A Coordinated Response at The Transcriptome and Interactome Level is Required to Ensure Uropathogenic Escherichia coli Survival during Bacteremia. Microorganisms. 2019; 7(9):292. https://doi.org/10.3390/microorganisms7090292 PMID: 31450662
- Reddick LE, Alto NM. Bacteria fighting back: how pathogens target and subvert the host innate immune system. Mol Cell. 2014; 54(2):321–8. https://doi.org/10.1016/j.molcel.2014.03.010 PMID: 24766896
- 42. Bhavsar AP, Guttman JA, Finlay BB. Manipulation of host-cell pathways by bacterial pathogens. Nature. 2007; 449(7164):827–34. https://doi.org/10.1038/nature06247 PMID: 17943119
- Itzhaki Z, Akiva E, Altuvia Y, Margalit H. Evolutionary conservation of domain-domain interactions. Genome Biol. 2006; 7(12):R125. https://doi.org/10.1186/gb-2006-7-12-r125 PMID: 17184549
- del Sol A, Carbonell P. The modular organization of domain structures: insights into protein-protein binding. PLoS Comput Biol. 2007; 3(12):e239. https://doi.org/10.1371/journal.pcbi.0030239 PMID: 18069884
- Elde NC, Child SJ, Geballe AP, Malik HS. Protein kinase R reveals an evolutionary model for defeating viral mimicry. Nature. 2009; 457(7228):485–9. https://doi.org/10.1038/nature07529 PMID: 19043403
- Rendon JM, Lang B, Tartaglia GG, Burgas MT. BacFITBase: a database to assess the relevance of bacterial genes during host infection. Nucleic Acids Res. 2019; 48(D1):D511–16.
- Lehar SM, Pillow T, Xu M, Staben L, Kajihara KK, Vandlen R, et al. Novel antibody-antibiotic conjugate eliminates intracellular S. aureus. Nature. 2015; 527(7578):323–8. <u>https://doi.org/10.1038/nature16057</u> PMID: 26536114
- Dyer MD, Neff C, Dufford M, Rivera CG, Shattuck D, Bassaganya-Riera J, et al. The human-bacterial pathogen protein interaction networks of Bacillus anthracis, Francisella tularensis, and Yersinia pestis. PLoS One. 2010; 5(8):e12089. https://doi.org/10.1371/journal.pone.0012089 PMID: 20711500
- Durmus Tekir S, Cakir T, Ardic E, Sayilirbas AS, Konuk G, Konuk M, et al. PHISTO: pathogen-host interaction search tool. Bioinformatics. 2013; 29(10):1357–8. https://doi.org/10.1093/bioinformatics/ btt137 PMID: 23515528
- Gerstein M. How representative are the known structures of the proteins in a complete genome? A comprehensive structural census. Fold Des. 1998; 3(6):497–512. <u>https://doi.org/10.1016/S1359-0278(98)</u> 00066-2 PMID: 9889159
- Szklarczyk D, Franceschini A, Wyder S, Forslund K, Heller D, Huerta-Cepas J, et al. STRING v10: protein-protein interaction networks, integrated over the tree of life. Nucleic Acids Res. 2015; 43(D1): D447–52. https://doi.org/10.1093/nar/gku1003 PMID: 25352553
- 52. Eichinger V, Nussbaumer T, Platzer A, Jehl MA, Arnold R, Rattei T. EffectiveDB—updates and novel features for a better annotation of bacterial secreted proteins and Type III, IV, VI secretion systems. Nucleic Acids Res. 2016; 44(D1):D669–74. https://doi.org/10.1093/nar/gkv1269 PMID: 26590402
- Palace SG, Proulx MK, Lu S, Baker RE, Goguen JD. Genome-wide mutant fitness profiling identifies nutritional requirements for optimal growth of Yersinia pestis in deep tissue. MBio. 2014; 5(4). <u>https:// doi.org/10.1128/mBio.01385-14 PMID: 25139902</u>
- 54. Deleage G, Roux B. An algorithm for protein secondary structure prediction based on class prediction. Protein Eng. 1987; 1(4):289–94. https://doi.org/10.1093/protein/1.4.289 PMID: 3508279
- Tartaglia GG, Pawar AP, Campioni S, Dobson CM, Chiti F, Vendruscolo M. Prediction of aggregationprone regions in structured proteins. J Mol Biol. 2008; 380(2):425–36. <u>https://doi.org/10.1016/j.jmb.</u> 2008.05.013 PMID: 18514226

- Campen A, Williams RM, Brown CJ, Meng J, Uversky VN, Dunker AK. TOP-IDP-scale: a new amino acid scale measuring propensity for intrinsic disorder. Protein Pept Lett. 2008; 15(9):956–63. <u>https://doi.org/10.2174/092986608785849164</u> PMID: 18991772
- Castello A, Fischer B, Frese CK, Horos R, Alleaume AM, Foehr S, et al. Comprehensive Identification of RNA-Binding Domains in Human Cells. Mol Cell. 2016; 63(4):696–710. <u>https://doi.org/10.1016/j.</u> molcel.2016.06.029 PMID: 27453046
- Grant BJ, Rodrigues AP, ElSawy KM, McCammon JA, Caves LS. Bio3d: an R package for the comparative analysis of protein structures. Bioinformatics. 2006; 22(21):2695–6. https://doi.org/10.1093/ bioinformatics/btl461 PMID: 16940322
- 59. Webb B, Sali A. Protein Structure Modeling with MODELLER. Methods Mol Biol. 2017; 1654:39–54. https://doi.org/10.1007/978-1-4939-7231-9_4 PMID: 28986782
- Ramirez-Aportela E, Lopez-Blanco JR, Chacon P. FRODOCK 2.0: fast protein-protein docking server. Bioinformatics. 2016; 32(15):2386–8. https://doi.org/10.1093/bioinformatics/btw141 PMID: 27153583
- Shannon P, Markiel A, Ozier O, Baliga NS, Wang JT, Ramage D, et al. Cytoscape: a software environment for integrated models of biomolecular interaction networks. Genome Res. 2003; 13(11):2498– 504. https://doi.org/10.1101/gr.1239303 PMID: 14597658